Author’s response to reviews

Title: Effects of a Parallel Arm Randomized Controlled Weight Loss Pilot Study on Biological and Psychosocial Parameters of Overweight and Obese Breast Cancer Survivors

Authors:

Susan Raatz (susan.raatz@ars.usda.gov)
Andrea Arikawa (andrea.arikawa@unf.edu)
Beth Kaufman (bkaufman@umn.edu)
Mindy Kurzer (mkurzer@umn.edu)

Version: 2 Date: 08 Jun 2017

Author’s response to reviews:

Dear Associate Editor,

We apologize for overlooking the guidelines for reporting randomized pilot trials. We have thoroughly reviewed the guidelines and, as a result, we have modified the manuscript considerably. We also changed the way the data were reported by removing p-values and using means and 95% confidence intervals, instead. All changes were made using the Track changes in word.

The changes made are highlighted on the uploaded Response to Reviewers.